Author's response to reviews

Title: Impact of a personal learning plan supported by an induction meeting on academic performance in undergraduate Obstetrics and Gynaecology: a cluster randomised controlled trial

Authors:

Richard P Deane (richarddeane@physicians.ie)
Deirdre J Murphy (MURPHYD4@tcd.ie)

Version: 4 Date: 23 February 2015

Author's response to reviews: see over
Dear Prof Azer,

Many thanks for your email from 20th February indicating that BMC Medical Education have accepted our study / manuscript for publication (‘Impact of a personal learning plan supported by an induction meeting on academic performance in undergraduate Obstetrics and Gynaecology: a cluster randomised controlled trial’). We are delighted to accept this offer for publication. We are particularly grateful to the reviewers for the most helpful and thorough peer review that has greatly enhanced the manuscript.

The attached revised version of the manuscript contains the following additional sections as requested: competing interests, authors’ contributions, authors’ information and acknowledgements.

Many thanks for agreeing to publish our study.

Yours sincerely,

Dr Richard Deane
Lecturer in Obstetrics & Gynecology,
Trinity College, University of Dublin.

Corresponding Author:
Dr Richard Deane,
TCD Department of Obstetrics & Gynecology,
Education Centre,
Coombe Women & Infants University Hospital,
Dublin 8,
Republic of Ireland.

e-mail: richarddeane@physicians.ie
Telephone: 00 353 1 4085365